Author's response to reviews

Title: Reliability of race assessment based on the race of the ascendants: a cross-sectional study

Authors:

Sandra C Fuchs (scfuchs@zaz.com.br)
Sylvia M Guimaraes (sylvia@cpovo.net)
Cristine Sortica (crisortica@conex.com.br)
Fernanda Wainberg (fernandawainberg@hotmail.com)
Karine O Dias (kodias@hcpa.ufrgs.br)
Mariana Uguini (marianacosta@via-rs.net)
Jose Augusto S Castro (joscsc@portoweb.com.br)
Flavio D Fuchs (ffuchs@hcpa.ufrgs.br)

Version: 5 Date: 15 Jan 2002

We are delighted in learning that our manuscript depends on minor edition to be published. We have changed it according to the request. Please, let me know if anything more is necessary. Best wishes, SF.